

Archive ouverte UNIGE

https://archive-ouverte.unige.ch

Article scientifique

Article 2021

Published version

Open Access

This is the published version of the publication, made available in accordance with the publisher's policy.

Disclosure to social network members among abortion-seeking women in low- and middle-income countries with restrictive access: a systematic review

Rossier, Clementine; Marchin, Angela; Kim, Caron; Ganatra, Bela

How to cite

ROSSIER, Clementine et al. Disclosure to social network members among abortion-seeking women in low- and middle-income countries with restrictive access: a systematic review. In: Reproductive Health, 2021, vol. 18, n° 1, p. 1–15. doi: 10.1186/s12978-021-01165-0

This publication URL:https://archive-ouverte.unige.ch/unige:154406Publication DOI:10.1186/s12978-021-01165-0

© The author(s). This work is licensed under a Other Open Access license <u>https://www.unige.ch/biblio/aou/fr/guide/info/references/licences/</u>

REVIEW

Open Access

Disclosure to social network members among abortion-seeking women in lowand middle-income countries with restrictive access: a systematic review



Clémentine Rossier^{1,2*}, Angela Marchin³, Caron Kim⁴ and Bela Ganatra⁴

Abstract

Background: Health care for stigmatized reproductive practices in low- and middle-income countries (LMICs) often remains illegal; when legal, it is often inadequate, difficult to find and / or stigmatizing, which results in women deferring care or turning to informal information sources and providers. Women seeking an induced abortion in LMICs often face obstacles of this kind, leading to unsafe abortions. A growing number of studies have shown that abortion seekers confide in social network members when searching for formal or informal care. However, results have been inconsistent; in some LMICs with restricted access to abortion services (restrictive LMICs), disclosure appears to be limited.

Main body: This systematic review aims to identify the degree of disclosure to social networks members in restrictive LMICs, and to explore the differences between women obtaining an informal medical abortion and other abortion seekers. This knowledge is potentially useful for designing interventions to improve information on safe abortion or for developing network-based data collection strategies. We searched Pubmed, POPLINE, AIMS, LILACS, IMSEAR, and WPRIM databases for peer-reviewed articles, published in any language from 2000 to 2018, concerning abortion information seeking, communication, networking and access to services in LMICs with restricted access to abortion services. We categorized settings into four types by possibility of anonymous access to abortion services and local abortion stigma: (1) anonymous access possible, hyper stigma (2) anonymous access possible, high stigma (3) non-anonymous access, high stigma (4) non-anonymous access, hyper stigma. We screened 4101 references, yielding 79 articles with data from 33 countries for data extraction. We found a few countries (or groups within countries) exemplifying the first and second types of setting, while most studies corresponded to the third type. The share of abortion seekers disclosing to network members increased across setting types, with no women disclosing to network members increased across setting types, with no women disclosing to network members beyond their intimate circle in Type 1 sites, a minority in Type 2 and a majority in Type 3. The informal use of medical abortion did not consistently modify disclosure to others.

Conclusion: Abortion-seeking women exhibit widely different levels of disclosure to their larger social network members across settings/social groups in restrictive LMICs depending on the availability of anonymous access to abortion information and services, and the level of stigma.

*Correspondence: clementine.rossier@unige.ch

¹ Université de Genève, Geneva, Switzerland

Full list of author information is available at the end of the article



© The Author(s) 2021. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Plain Language summary

Women seeking an induced abortion in LMICs often face inexistent or inadequate, difficult to find and/ or stigmatizing legal services, leading to the use of informal methods and providers, and unsafe abortions. A growing number of studies have shown that abortion seekers contact social network members beyond their intimate circle when seeking care. However, results have been inconsistent. We searched Pubmed, POPLINE, AIMS, LILACS, IMSEAR, and WPRIM databases for peer-reviewed articles published in any language from 2000 to 2018, concerning abortion information seeking, communication, networking and access to services in restrictive LMICs. We screened 4101 references, yielding 79 articles with data from 33 countries for extraction. We grouped countries (or social groups within countries) into four types of settings: (1) anonymous access possible, hyper stigma; (2) anonymous access possible, high stigma; (3) non-anonymous access, high stigma; (4) non-anonymous access, hyper stigma. Most studies fitted Type 3. Disclosing to network members increased across setting types: no women confided in network members in Type 1 settings, a minority in Type 2 and a majority in Type 3. No setting fitted Type 4. The informal use of medical abortion did not modify disclosure to others. Abortion seekers in restrictive LMICs frequently contact their social network in some settings/groups but less frequently in others, depending on the availability of anonymous access to abortion care and the level of stigma. This knowledge is useful for designing interventions to improve information on safe abortion and for developing network-based data collection strategies.

Keywords: Unsafe abortion, Social network, Access to care, Low and middle income countries

Background

Numerous studies have looked at the links between interpersonal ties and health outcomes [1]. Such effects are both positive and negative, and operate through several pathways: [2] first, emotional support, conflict and practical help directly affect individuals' health resources; support is usually provided by family members or close ties. Second, social norms influence and control individual aspirations and health behaviours, and new normative and behavioural models are acquired through social learning. Third, information and socioeconomic resources are swapped between members of groups sharing similar interests. In the field of sexual and reproductive health, the first two mechanisms-social support and social influence/learning-have attracted attention, because financial costs and sociocultural opposition remain fundamental obstacles to reproductive healthseeking behaviours [3]. For instance, financial transfers within women's close networks matter for their access to prenatal care in Mali [4]. Fears of side effects, fuelled by rumours circulating in social networks reflecting social influence/social learning processes [5] are also identified today as a major obstacles to modern contraceptive access [6].

While a staple of stratification studies, the exchange of strategic information and resources among fellow group members with similar interests—corresponding to Bourdieu's definition of social capital [2]—is less often studied in reproductive health. Yet this mechanism is arguably crucial in women's ability to access (informal) care when engaging in stigmatized sexual behaviours or when experiencing stigmatized health problems (nonmarital sexuality, abortion, sexually transmitted diseases, etc.) Adolescents in low-income countries, for example, whose sexuality is a social taboo, turn to their peers for information and advice that is difficult to obtain from authorized sources, and this has prompted the development of peer-led interventions [7]. This pattern reflects the fact that health care for stigmatized reproductive practices in low and middle income countries (LMICs) either remains illegal, as is frequently the case for induced abortion [8], or, when legal, is often inadequate, difficult to find by its target group and/ or stigmatizing, which results in women deferring care or turning to informal information sources and providers [9].

For women seeking an induced abortion in LMICs, access to care is often restricted, leading to unsafe abortions; each year, 97% of the estimated 25.1 million unsafe abortions worldwide occur in the developing world [10]. Unsafe abortion leads to increased health care costs and morbidity, and is a leading cause of preventable maternal mortality, causing at least 22,000 deaths each year [11]. Just half (50.5%) of all estimated abortions in developing countries are safe, compared with 88% of those in more developed countries, a proportion which takes into account issues of underreporting and non-diagnosis [10]. In developed countries, less safe abortions are linked to the use of outdated methods (dilatation and curettage) rather than problems of access: in countries with restrictive laws, women are often able to access services with the help of telemedicine or feminist groups, or by travelling abroad [12]. In developing countries, the expanding informal use of medical abortion (self-administration of mifepristone) in the three last decades has decreased the number of "least safe" abortions in favour of "less safe" abortions [10]. "Least safe" abortions are performed

by untrained providers using folk methods (potions, sticks,..); "less safe" abortions correspond to women's self-administration of mifepristone or use of dilatation and curettage by trained providers; "safe" abortions are conducted with modern clinical/ surgical methods (vacuum aspiration) or medical means by medically trained providers.

In the case of abortion, informal advice tends to be unhelpful because the diverse folk recipes which circulate in social networks are generally ineffective [12]; women mainly succeed in obtaining an abortion by visiting an underground or informal abortionist/drug seller whoby definition-is not of public knowledge. So women and their partners (or mothers) turn to their social network to "search for an abortionist", as already shown in a seminal study conducted in the United States before abortion was legalized [13]. To obtain the information they need, women and couples have to *disclose* their situation to their network contacts, to one contact at least; this contact can then continue the search on behalf of an (unspecified) friend [14]. Disclosure of stigmatized behaviours is possible under two conditions: with trusted others (close ties, or weaker ties in the same interest group, such as young people or women) [15] or when the transgression is common to both parties (as with the abortion provider or another abortion seeker) [16].

A growing number of studies have shown that abortion seekers confide in their social network members in order to find informal or illegal abortion services in LMICS where access to abortion services is restricted (i.e. where illegal or informal providers are active). However, results have been inconsistent. In some restrictive LMIC settings, or for some groups, disclosure seems to be very limited, perhaps due to very high levels of stigma [17]. In other settings, especially countries with a mix of formal and formal providers such as India, disclosure to social network members seems to be relatively infrequent [18], for reasons which remain unclear. Moreover, it is unclear how the spread of informal medical abortion in restrictive LMICs has affected disclosure to network members, as the medications and information on how to use them may be easier to find than other types of providers and methods.

This systematic review aims both to identify the restrictive LMIC settings in which women and their partners confide in members of their social networks to obtain information about abortion methods and providers, and to explore possible differences or similarities in information sources between women seeking informal medical abortion and those seeking other abortion methods. As detailed below, we categorize settings into four types by possibility of anonymous (i.e. confidential) access to abortion services and local abortion stigma: (1) anonymous access possible, hyper stigma (2) anonymous access possible, high stigma (3) non-anonymous access, high stigma (4) non-anonymous access, hyper stigma. Studying disclosure to social networks may shed light on the circumstances in which women seek different methods of abortion—safe or less safe—in LMICs with restrictive access, and provide useful knowledge for designing interventions to improve access to information and for developing network-based data collection strategies.

Methods

We searched Pubmed, POPLINE, AIMS, LILACS, IMSEAR, and WPRIM databases and performed handsearching for peer-reviewed articles published in any language from 2000 to November 2018 concerning disclosure to network members among abortion seeking women in LMICs where access to safe abortion is restricted. Disclosure to the conjugal or parental dyads, and processes of social support and social influence are outside the scope of the present analysis. Rather, we focus on information sharing across social networks beyond their intimate circle in line with Bourdieu's definition of social capital-the sharing of information and resources in larger groups with common interests and identities[2]-to find abortion providers or methods. An appropriate definition of social network membership in this context [19] may be the people one knows by name, has spoken to in the last year, and could contact if desired [20]. At the other end of the relational continuum, the health professionals that women talk to (school psychologists, health staff at abortion clinics, etc.) are not considered as "social network members," unless they already belong to the women's or the couples' sets of social relations.

In practice, we searched for all LMICs, and removed manually the studies concerning settings with unrestricted access to abortion such as Vietnam or Mexico City. Note that two studies concerning residents of restrictive LMICs travelling to another country with nonrestrictive access (Thailand, the United States) were also included, as the abortion seekers relied on their social network to organize their travel and access to care. For all these countries, observational and qualitative studies including women who underwent a medical or surgical abortion (the exact surgical method is not always specified) or an abortion by other methods at any gestational age and which mentioned communication/networking, information seeking or broader terms such as "experiences" or "processes" in the titles/abstracts were screened for inclusion (See Additional file 1 for the search terms).

We reviewed titles and abstracts, and the full article, when necessary, to identify studies for inclusion. Descriptive and qualitative studies were of particular interest, as well as studies using snowball or respondent-driven sampling methods, as these methods imply the existence of social networking among abortion seekers. Interventional studies were considered when attention was given to information seeking, communication, or social support beyond intimate ties.

Study quality was determined using standardized checklists for each study type, and rated from 1 to 5 based on a set number of criteria (different for qualitative and quantitative studies). The most commonly encountered problem was that of missing information on the methods used for qualitative analysis (content analysis should have been mentioned), on data collection or data analysis (such as year missing, demographic data not presented, etc.); in some cases, discussion of biases and limitations was also missing. Studies were downgraded for each problem encountered. A minimum quality level of 3 was set for inclusion in the analysis, corresponding to a level of information sufficient to identify the data biases (for example, only abortion complication patients) so that the review authors could consider it in the analysis even when not mentioned by the study authors.

We did not compute summary measures of association due to heterogeneity in study designs, populations, and outcome measures. We extracted data from the selected studies, initially separating qualitative from quantitative data. More precisely, we extracted all the quotes that referred to how abortion seekers talk to others about abortion in general or disclose to others about this event (to whom, under what conditions, how frequently, for what reasons) and how women who had an abortion are contacted by other abortion seekers. We also extracted quotes about how women get information on abortion services and methods.

We noted in addition all information about the context of abortion for the groups of individuals studied. We documented the following contextual dimensions when the relevant information was provided in the articles:

- Legal status of abortion at the country level;
- Share of illegal abortions;
- Whether abortion services (legal or not) can be accessed confidentially by women on their own, without the help of their community or network (i.e. via flyers or posters, helpline, Internet, or professionals such as pharmacists or healthcare providers, teachers or school psychologists, feminist associations, etc.);
- Level of stigma attached to abortion (mistreatment during abortion or post-abortion care, fear of judgment by others, self-judgement, reaction upon disclosure, etc.)

Based on these dimensions and on the previous literature, we constructed a typology of settings based on two intersecting dimensions: possibility of anonymous (i.e. confidential) access to abortion services and level of stigma. For the first dimension we distinguished settings where anonymous access to services is possible for at least a share of abortion seekers (whether or not these services are legal or safe) versus settings with no anonymous access. For the second dimension we distinguished between hyper stigma and high stigma, hyper stigma being characterized by rejection and hostility on the part of close ties and health staff in response to disclosure; high stigma being characterized by strong disapproval of the practice in general but a relative understanding on the part of close ties and health staff. These two dimensions, when intersected, yielded four possible types of settings:

- 1) (A share of the) women can access services anonymously but cannot disclose to contacts (hyper stigma)
- 2) (A share of the) women can access services anonymously and can confide in trusted others in order to find services (high stigma)
- 3) Women overwhelmingly need help from personal contacts to find services and can confide in trusted contacts in order to find services (high stigma)
- 4) Women overwhelmingly need help from personal contacts to find services but cannot disclose to contacts (hyper stigma).

In a final step of the analysis, we pooled the extracted content into one description per country. For countries with several studies we sometimes distinguished between different groups (married and not married, immigrants), different regions or time periods when women's disclosure behaviours or access to abortion services were different. We placed each country/group in one of the four contextual categories, and analysed common patterns of disclosure to network members across types, distinguishing abortion seekers using informal medical abortion from those using other methods.

Results

The systematic search of Pubmed, POPLINE, and LILACS (see Additional file 1 for search terms), yielded 4101 references, 79 of which were included for analysis (Fig. 1). The AIMS and WPRIM databases were screened by hand, and yielded no additional articles. The IMSEAR database was non-functional during the time of study screening.

The 79 studies selected for inclusion took place in 33 different countries in sub-Saharan Africa (15 countries),



Asia [8], Latin America [7], and the Middle East [3], or concerned residents of those countries who sought abortions abroad (Table 1).

Over half of these studies [21] had been published in the previous five years (2014–2018), the rest between 2001 and 2013. More than half of them [22] used qualitative methods (ethnography, in-depth interviews, focus groups) and the rest were based on quantitative analysis of data collected in representative samples (communitylevel surveys or abortion care patients) or followed a mixed-methods approach (quantitative data from a survey of abortion care patients supplemented by in-depth interviews). All 79 studies met the minimum quality requirement set at 3 out of 5.

We classified all the populations (countries/groups) studied in this meta-analysis into one of the first three configurations of settings. We found no study exemplifying the fourth type of setting (non-anonymous access and hyper stigma), probably because abortions are, by definition, rare in such settings. In these settings women with unplanned pregnancies, and/or their partner or mother, cannot access an anonymous abortion provider since such providers—typically located in a health centre and advertising through flyers or the internet—do not exist; neither can they turn to underground abortion providers, which are probably rare as they need word-of-mouth referral to thrive (and women who want an abortion or have had one do not dare disclose to others because of hyper stigma). In such settings, women with unintended pregnancies likely carry them to term, despite the potential adverse consequences (Table 2).

Type 1: Anonymous access and hyper stigma

A first configuration of situations concerns countries/ groups where abortion seekers speak to no one about what they are going through, except the abortion provider and sometimes a partner or parent (i.e. an intimate tie). They disclose to no one in their larger social network, because of the severe stigma attached to being pregnant /having an abortion in their situation: "family honour," "suicide of parents," and "killing a baby," are terms mentioned by the respondents in the qualitative studies. These women all access (legal or illegal) abortion services anonymously, as services can be approached confidentially.

A qualitative study of *unmarried women in Addis Ababa, Ethiopia* (post-abortion care patients) describes how these women keep their pregnancy secret because of the shame it would bring to themselves and their family. They go to legal abortion clinics in other parts of the city, and if they cannot afford the prices charged, are approached on the parking lot by illegal providers

Table 1: 79 studies in 33 countries

Region/ countries	Studies
Africa	
Ethiopia	Kebede [24], Kebede [23], Alemayehu [46]
South Africa	Orner [38], Orner [37], Gerdts [39], Harries [40]
Nigeria	Bankole [94], Alubo [93]
Zambia	Coast [59], Freeman [60], Cresswell [61], Dahlback [62]
Kenya	lzugbara [63] Osur [64] Mitchell [66] lzugbara [65], Jayaweera [48] Penfold [67]
Ghana	Hill [68] Kumi-Kyereme [69] Aniteye [70], Appiah-Agyekum [71], Rominski [73] Ganle [72], Rominski [74]
Uganda	Nyanzi [75], Atuyambe [76], Marlow [77]
Tanzania	Norris [78], Plummer [79], Rasch [80]
DRC	Rouhani [81], Burkhardt [82]
Burkina Faso	Ouédraogo [83], Rossier [14]
Benin	Baxerres [84] (also Burkina Faso)
Madagascar	Pourette [85]
Botswana	Smith [86]
Malawi	Jackson [95]
Gabon	Hess [96]
Latin America	
Colombia	Brack [25], DePineres [26]
Bolivia	Bury [47]
Brazil	Silveira [22], Arilha [51], Diniz [52], Madeiro [53], Nunes 2013, Heilborn [54]
Chile	Casas [55], Palma Manriquez [56]
Argentina	Ramos [57]
Mexico/California	Grossman [91]
Haiti	Albuja [97]
Asia	
India	Kalyanwala [27], Jeejeeboy [28], Banerjee [29], Banerjee [32], Banerjee [33], Elul [30], Behera [31], Baner- jee 2015, Elul [35]
Nepal	Puri [43], Rocca [44], Ohja [41], Andersen [42]
Bangladesh	Messinger [45]
Sri Lanka	Arambepola [50]
Malaysia	Tong 2012
Philippine	Gipson [87]
Cambodia	Petitet [21]
Burma/Thaïland	Arnott [88], Tousaw [89], Tousaw [90]
Middle East	
Iran	Zamanian [36]
Saudi Arabia	Alsibiani [49]
Palestinian OT	Shahawy [98]

operating nearby. Only one young woman reached out to a community member; she asked a prostitute in her neighbourhood for advice on where to get an abortion [23, 24].

Another qualitative study in *Colombia* describes a set of women who obtained a legal abortion from one of the few legal clinics operating in the capital; these women did not dare speak about their pregnancy and abortion with others because of the very strong anti-abortion stigma. Instead, they used the Internet or

referral from community psychologists or teachers, and only occasionally a friend, to find out about the clinics [25]. Another study in Colombia shows that women use the Internet to find addresses of clinics or places to buy misoprostol, and they are approached by illegal providers waiting outside the clinics or sales venues they visit in person. Talking to friends is mentioned with respect to getting information about misoprostol, but it is not clear whether women disclose their pregnancy to obtain it. The women always appear to have gone

Types	Studies
Anonymous access possible (for a share of women), hyper stigma	Kebede [24], Kebede [23] Brack [25], DePineres [26]
Anonymous access possible (for a share of women), high stigma	Kalyanwala [27], Jeejeeboy [28], Banerjee [29], Banerjee [32], Banerjee [33], Elul [30], Behera [31], Baner- jee 2015, Elul [35] Zamanian [36] Orner [38], Orner [37], Gerdts [39], Harries [40] Rocca [44], Ohja [41], Andersen [42], Puri [43] Alemayehu [46] Bury [47] Messinger [45] Petitet [21]
No anonymous access, high stigma	Alsibiani [49] Arambepola [50] Silveira [22], Arilha [51], Diniz [52], Madeiro [53], Nunes 2013, Heilborn [54] Casas [55], Palma Manriquez [56] Ramos [57] Coast [59], Freeman [60], Cresswell [61], Dahlback [62] Izugbara [63] Osur [64], Mitchell [66], Izugbara [65], Jayaweera [48], Penfold [67] Hill [68], Kumi-Kyereme [69], Aniteye [70], Appiah-Agyekum [71], Rominski [73] Ganle [72], Rominski [74] Nyanzi [75], Atuyambe [76], Marlow [77] Norris [78], Plummer [79], Rasch [80] Rouhani [81], Burkhardt [82] Ouédraogo [83], Rossier [14], Baxerres [84] Pourette [85] Smith [86] Tong 2012 Bankole [94], Alubo [93] Jackson [95] Hess [96] Grossman [91] Albuja [97] Arnott [88], Tousaw [89], Tousaw [90] Shahawy [98] Gipson (87)
No anonymous access, hyper stigma	No studies

 Table 2
 79 studies in four types of settings classed by criteria of anonymous access and stigma

through the abortion process alone (only one mentions having told her parents) [26].

A qualitative study of *unmarried women in India* (states of Bihar and Jharkhand) who had an abortion (and self-reported it in a survey) shows a similar pattern. Most of these women informed their partner and were supported by him. Some received practical support from an older family member, although he/she tended to respond with hostility, even if the pregnancy was the result of forced sex. Those who told neither their partner nor their family went through the abortion on their own. All these young women used legal or informal services without the help of their social network, because pregnancies outside marriage are strongly condemned in that setting, and services can be found anonymously [27, 28].

The women described in these studies managed to obtain an abortion on their own or solely with the help of a significant other, such as a partner or parent. It seems, therefore, that in these settings women will only have an abortion if they are able to access abortion services without disclosing their situation to other social contacts.

Type 2: Anonymous access for a share of the women and high stigma

This second type includes settings where abortion seekers only sometimes disclose their secret to trusted friends, relatives, or members of their social group. Two features of the cultural and institutional context seem to explain this pattern of infrequent disclosure. First, abortion is a stigmatized event and women prefer to keep this experience very private if possible. Second, abortion services (whether illegal or legal) are publicly advertised, although not perfectly; only when women cannot find a service provider or method by themselves do they reach out (secretly) to trustworthy or potentially knowledgeable social network members.

Several studies of *married women in India* show that women start by discussing their pregnancy with their husbands and sometimes their in-laws. Most women (or their intimate ties) know where to find legal abortion services. For example, half of the women in a community sample can cite a correct source [29], and a qualitative study mentions that women overwhelmingly know where

to find legal services [30]. However, these legal services are often deemed expensive or otherwise inaccessible. Many trained and untrained providers offer illegal services at the local level, and these providers are directly solicited by abortion seekers to obtain information. In one study reviewed, 89% of women who had an abortion in a rural community got their information from local health providers [29], in another study in the slums of Mumbai, 69% of women who had an abortion obtained information on the procedure from local health staff [31]. Women or their closest dyad (partner, in-laws) turn to friends/other community members only when they are unable themselves to locate the cheaper and more convenient alternatives to legal options. As a result, most abortion seekers do not talk to a friend or community member (in a study of women with abortion complications, only 1 in 5 got advice from a friend) [32-34]. In a community survey of women in the slums of Mumbai, only 19% received help from friends or other women who had had an abortion in the community [31]. For those in need, the help of friends, although only occasional, is nevertheless instrumental in getting an abortion. Another quantitative study in India showed that abortion seekers seldom talk to friends, but that having friends to confide in increases the probability of obtaining an abortion [35]. While sufficient information is lacking to classify the study, married women in Iran could be in a similar situation as those in India (relatively easy access to medical staff and limited disclosure to networks), as information about abortion seldom extends beyond the intimate circle [36].

The situation in *South Africa* appears to be analogous. One qualitative study of HIV-positive women who had had a legal abortion describes how abortion services were familiar to many of them. "There was general awareness that free abortions were available in public health sector facilities, gained mostly from local media and clinic posters, but also from knowing other women who had used the services" [37]. (p. 787) (see also [38]). The study underlines women's reluctance to talk to others about their abortion (a taboo much stronger than that of their HIV status), because of the high level of stigma attached to this practice. Another study in South Africa reports that women are relatively knowledgeable about legal abortion services, but fearing the exposure and stigma linked to using legal services they prefer to find alternative solutions [39]. Study participants (all informal sector abortion service users) were found through a respondent-driven-sample (RDS): "44% of participants reported seeking informal sector abortions because they worried someone would find out about the abortion, and 30% reported concerns about mistreatment and stigma from providers in the formal sector"(p.5). Knowledge of these alternative solutions is obtained through friends mainly (63%), but also through posters or flyers (23%). In the group of abortion seekers studied (all informal sector abortions), most women shared their abortion experience with others and knew others who had had an abortion. Similar results were found for women who had had a delayed second-trimester abortion in the public sector [40]. These informal sector or late public-sector experiences contrast with those described by legal abortion seekers in South Africa; on the whole, only a fraction of the women concerned disclose to social networks.

In *Nepal*, abortion became legal relatively recently, but many women still use informal and unsafe services [41–43]. Accordingly, some women still discuss their pregnancy and seek abortion advice from friends or community members. In a quantitative study, 20.3% of women told a friend they were getting an abortion, 25.8% found out how to get an abortion from a friend, and 11.4% reported that a friend was a source of support during the abortion process [44].

Studies in *Bangladesh*, [45] *Ethiopia* [46] (on postabortion care for patients of any marital status), *Cambodia* [21] and *Bolivia*, [47] while providing limited information on disclosure to network members, seem to indicate similar situations. They suggest that while many women know about and can access abortion services anonymously, not all women want these services or know about them, so turn to friends and community members for information on underground services.

Type 3: Non-anonymous access and high stigma

In the third set of situations, by far most the most common in the articles analysed, women seeking an abortion, and sometimes their partners/ mothers, overwhelmingly confide in members of their social networks under the cover of secrecy to obtain information about abortion services. Two features of the context seem to explain this pattern. First, abortion is stigmatized, but friends or selected knowledgeable community members are trusted to keep a secret. Second, information about abortion services is not publicly available (whatever the status of the service, legal or not), and health providers who can be approached anonymously do not refer abortion seekers to abortion services. Note that in these settings, women sometimes begin by trying well-known traditional or folk medicine methods on their own. However, these attempts are often unsuccessful, or end in complications that reveal the abortion to many more: for example in the Nairobi slums: "One participant said, 'I know of a friend who took some Quencher [fruit flavoured drink] that was not diluted and then she bled excessively; by the time she was getting to the hospital, she was dead" [48]

(p. 6). Women thus usually need to talk to trusted others to locate an effective method or provider.

Most studies (and countries) described in these articles are in this category. Since women need to talk to trusted others to locate abortion providers or effective methods, and since abortion is stigmatized but accepted by network members, a large majority reach out to their social networks. The only women who do not are those who can access services directly from a health provider (for example their boyfriend is a medical student) and those who have no one in their social network they can trust. The latter can still self-induce the abortion using traditional folk methods, although they are often ineffective and dangerous, as mentioned above. Except in these two situations, trusted network members are consulted to locate "knowledgeable others" (people who know about a method or a provider). Who these "knowledgeable others" are exactly varies across settings, but women who had an abortion recently are mentioned in most studies. Other resourceful people cited are prostitutes, illegal drug sellers, feminist groups, people bringing clients to illegal abortion providers, and taxi drivers. Sometimes these informants are approached directly, even if they do not belong to the seeker's social network; however, this manoeuvre is risky and it is safer when a network member acts as an intermediary.

Under this pattern of information-gathering, a significant share of women in a community will have heard of a network member who has had an abortion, because they have been contacted by them to locate a method or provider. In Saudi Arabia, 82% of misoprostol users knew of another user [49]. In Sri Lanka, "the majority of [abortion seekers] approached their partners and/or immediate associates to obtain more information about the persons/places available for pregnancy termination (60%) and to accompany them to abortionists (52%)". According to this study, in the absence of information in their social networks, women in Sri Lanka relied on "unknown sources such as taxi drivers" [50]. In Brazil, several studies indicate that women rely on their network to locate where misoprostol can be procured. Who can be reached through these networks is very important, and can change the experience entirely: [22] members of feminist groups were mentioned as especially helpful in this regard, as well as illegal drug sellers, and women who had used misoprostol themselves [51]. "The speed and ease with which a woman triggers a wide network of care and facilities for an abortion is one of the signs of how the culture of abortion is shared among women in Brazil... [Women who had an abortion] assist other women to have an abortion" [52] (p. 1679, translated from Portuguese). A study on Brazilian sex workers also reports that abortion seekers get information on misoprostol from other sex workers, sometimes directly from drug sellers [53]. In another study in Brazil, 30 teens having post-abortion care were interviewed: 80% got their information on where to buy Cytotec from friends; they consulted the Internet to find out *how* to take the drug. Note that Misoprostol is not the main method everywhere in Brazil, and in some cases, economically advantaged women do not need the help of their networks to get safe (albeit illegal) abortions from abortion clinics [54].

In Chile, most abortion seekers also appear to rely on their friends to locate the abortion drug, but some women manage to find misoprostol on the Internet by themselves: "The sellers are normally searched for online or through a friend or acquaintance who previously bought the pills or knows a seller... A participant who had an abortion five years prior to the interview now volunteers to help and support other women" [55]. A study on a sample of 30 university students obtained by snowball sampling states: "Medical abortion [...] was common knowledge among these young women, who directly or indirectly knew of other young women who had used it. This knowledge circulates within young women's circles, despite being 'forbidden'". Friends connect abortion seekers to health providers or feminist groups for exact information on where to obtain and how to use drugs for medical abortion [56].

In a qualitative study of women who had used medical abortion in *Argentina*, networking was very important in deciding to proceed and in obtaining the necessary medication. While the Internet and feminist groups helped (two women bought their product directly on the Internet), the fact that a prescription was needed made things difficult. "The complexity of this endeavour depended on getting the prescription needed to purchase it; having contacts who could identify which pharmacy to go to; finding a pharmacy that would sell it; having enough money; and getting the support of friends and/or family"[57] (p.7).

Although abortion is legal to a certain extent in Zambia (more precisely, abortion policies are ambiguous and the provision of legal abortion care is low) [58], research indicates that women obtain legal abortion services largely thanks to the help of their network. A few women who have no one to confide in end up having the most dangerous abortions and make many attempts with improbable methods; conversely, women who happen to know a health provider get very rapid access to safe abortion [59, 60]. Many women thus know about someone who has had an abortion: in a random sample of women "21% knew of at least one person who had had an induced abortion over the past 5 years" [61]. (p.3) In a study of young Zambian women presenting to a hospital with abortion complications, some quotes indicate that girls were guided by friends to a method, but other women seem to resort to self-administered folk methods to avoid talking to others, with their attempts ending in complications [62].

In Kenya, while a few hospitals offer safe, legal procedures, they are expensive, and women fear being stigmatized and exposed to the public eye if they use them. Consulting friends can be helpful for obtaining an anonymous abortion, as is illustrated by the following quote: "One respondent offered: 'I went to a Traditional Birth Attendant because she had helped some people I know and she keeps secrets. I did not even know she provides abortion services to women. It was a friend that she helped who directed me to her" [63] (p.14). Another study undertaken in Kenya (with interviews of illegal providers) shows how former abortion clients are key links to future clients for abortion providers: "A clandestine abortion provider indicates: 'I get at least three cases per month, mostly one per week. They are mostly school girls. They are referred by those I have treated before'" [64] (p.36). The study also confirms that prospective abortion seekers talk to a large range of people in their network before securing access to abortion (see also [48, 65]). For example, in a study of 614 students in Nairobi, almost half (45%) of the young people in the study knew peers who had had abortions [66]. But in a study with 22 women who attended six private clinics (abortion and post-abortion care) offering legal abortions in Western Kenya, only some of the women relied on friends to find methods; others were referred directly to the clinics by health staff or pharmacists [67]. These parts of Kenya are more similar to type 2.

Like Kenya, Ghana is a country where safe, legal abortion services have been difficult to access for a long time, so many women relied heavily on friends until recently. "Friends were the people most often informed about planned abortions and were contacted because they may know of a good and inexpensive method, are trusted to keep the abortion a secret, and to help if anything goes wrong"[68] (p. 2019–20). Moreover, abortion seekers in Ghana appear to consult quite a few people even before deciding to terminate, when they are unsure what to do about the pregnancy [69]. Recently, some legal abortion services have become available in Ghana, so some populations living near these services can be labelled as type 2, as some women get an abortion directly without telling others [70-72]. In another study we see that women using medical abortion informally do so with the help of their networks solicited in secret, although this does not seem to be the case for women using legal services [73]. In a study of university students, 42% knew someone who had had an abortion [74].

An in-depth qualitative study of young taxi drivers in *Uganda* sheds light on how they help women and couples get abortions, and on how younger people in general help each other to circumvent the strict demands of the older generations and the authorities with regard to morality and sexual conduct [75]. Another study on adolescents in Uganda shows that young people know about others who are seeking an abortion since they talk about them in focus groups [76]. In a study on sex workers in Kampala, all of the them consulted their fellow sex workers, friends, peer educators or community outreach educators when deciding how and where to obtain an abortion [77].

In a study in Tanzania, on the Island of Zanzibar, researchers were very successful in implementing a RDS strategy for abortion seekers "Despite the stigma of abortion, women talk with social contacts about abortion, and thus can connect each other via chain-referral sampling" [78] (p.7). Investigators were also able to get many detailed accounts of the abortion experience of friends in a number of villages in mainland Tanzania [79] (see also [80]). In the DRC likewise, RDS was used to interview a sample of women survivors of war and sexual violence, some of whom terminated a pregnancy that resulted from rape. The study shows that for these women, social networks were valuable in finding an abortion method. The snowball sampling strategy also suggests that women are aware of others in their networks who have undergone abortion [81, 82].

In Burkina Faso, a study provides qualitative evidence of such mechanisms and uses it to collect quantitative data on abortions among the respondents' trusted others [14]. A qualitative study in the same country confirms the key role of friends in securing abortion services, although one resourceful woman (a law student) obtained misoprostol directly from health personnel by arguing her case [83] (see also [84]). These mechanisms are confirmed for Benin, where friends or relatives are always used to locate one of the numerous small private health centres far from home where the abortion can be done discreetly [84]. Women in Madagascar, younger women especially, consult their friends about abortion. They also consult their network of neighbours/personal contacts who are medical professionals, either when they decide to terminate a pregnancy or if complications arise [85]. In Botswana, a qualitative study found that "most of the providers are women, and the process of finding an illegal practitioner and accessing the service operates through a women-only system of information: 'you'll always find someone who knows someone who can help you["][86] (p. 46). This is also the case in Malaysia [86] and in the Philippines [87].

A study of border residents of *Myanmar* (where abortion is strongly prohibited) who travelled to Thailand (where access to abortion services is legal on health grounds) depicts a similar context. Not all of these women can access legal services on their own and some go to informal providers on either side of the border with or without the help of friends [88]. Burmese migrants also need their friends to navigate the complex and unfamiliar Thai system of legal abortion [89, 90]. Similarly, in a study of 17 *Mexican* women who crossed the border to use legal abortion services in California, the women reported learning about the clinic from family members, friends, or the Internet. In some cases, they or their partners had previous experience of these services [91].

The remaining studies provide less detailed results, but similar situations appear to exist in a number of other countries, as in Malysia [92]. In Nigeria, older studies, while fragmentary, depict a situation where most women disclose to others in order to find abortion methods or providers [93, 94]. The situation today may be closer to type 2, as in Ghana or Kenya. The situation appears to be similar for *Malawi* (safe abortion for those with good connections) [95]. In Gabon, a small study indicates that in most cases women turn to friends to locate a method or provider [96]. In *Haiti*, providers operate clandestinely and the topic remains secret, but focus group participants mention friends who have had abortions [97]. In the Palestinian Occupied Territories, 4 out of 10 respondents (40 total, general populations) know a friend who has had an abortion. A range of options to terminate exist but seem difficult to access [98].

Discussion

The likelihood for women of using their networks and confiding in trusted others to locate abortion providers appears to be related to the legal status of abortion. Settings where women can access abortion services anonymously (and thus do not need to talk to their friends) (Types 1 and 2) are usually also those where abortions are allowed, on at least some grounds. But not all settings where legal abortion exists offer "anonymous" access to these services, Zambia being a striking example (Type 3). Moreover, in countries where abortion is legal, many or most women may refuse to use these services, deemed as too costly or too far away, not sufficiently confidential, or stigmatizing, as in South Africa or India, and more recently in Kenya or Ghana (Type 2). A varying share of women (but not the majority) consult their networks for information on alternative services in these countries. On the other hand, countries where abortion is illegal and safe services are unavailable, anonymous access to services may still exist thanks to the Internet, as is the case in Bolivia, which is similar in this respect to high-income restrictive countries.

Informal self-use of medical abortion does not seem to be systematically associated with more or less information sharing. In some cases, like in Bolivia (also documented in Chile), information on the Internet is sufficient to obtain and self-administer this method, so abortion seekers can avoid asking their networks for help (as in high-income countries with restrictive abortion laws). But in other places, like India or the Philippines, using this method informally requires more assistance from trusted network members and knowledgeable others, because women do not know where to buy the drug and how take it, unlike women who go to a publicly known (albeit often also informal) surgical abortion provider.

Another interesting finding is that disclosing to trusted network members is not sufficient to find abortion services directly: these contacts often go on to locate knowledgeable informants in the community. These resourceful informants are women who have had an abortion themselves, or health workers. In some settings, sex workers, taxi drivers, illegal drug sellers, people working with illegal providers and members of feminist groups are key informants in the search for an abortion provider. Part of the information sharing will converge towards these individuals. They are not part of women's networks, but have usually helped a network member (or a friend of a network member) in the past.

This study has a number of strengths. Having multiple team members to perform the data analysis contributed to internal validity. Multilingual team members also improved the access to data published in languages other than English. A limitation of this study is that much of the data is based on self-reported information, and is subject to bias, especially given the sensitive nature of the topic. Moreover, there is no recognized model for collecting data on information-sharing or disclosure behaviours of abortion seekers, so the studies reviewed here seldom collected enough data. They all document the sharing of abortion information between abortion seekers and their networks. While disclosure can be assumed in most cases, this is not always explicitly stated. When disclosure is explicitly documented, these studies usually distinguish the person spoken to (partner, mother or other relative, friends, health personnel), but precision is sometimes lacking. The exact type of information shared is not always clear either. Similarly, the exact conditions under which disclosure is taking place (degree of trust, secrecy) is rarely specified. While the incomplete nature of the information on networking in these studies is a real limitation to our classification, the fact that countries were often covered in multiple studies helped construct

a typology of disclosure for countries or groups within countries.

Comprehensive data collection on the topic could include the following topics: (1) The extent to which information on access to abortion services is public (i.e. how confidentially abortion services can be accessed: ads, Internet, health providers as a source of information) and how acceptable publicly available options are to women (stigmatization during care, privacy, costs). (2) The degree of abortion stigma: will close friends and relatives be understanding when told in secret about a pregnancy to be aborted? (3) General information about abortion overheard or obtained from others should be distinguished from disclosure to others to obtain more precise information. (2) In case of disclosure, with whom precisely the abortion was discussed should be documented: partner, mother/father or tutor or mother/father in-law; friends; other relatives (cousins etc.); other network members (specify condition of trust). (4) Whether these network members had an abortion themselves should also be documented. (5) Whether these network members helped in accessing services directly or whether they had contacted someone else, and whom (specify conditions of trust).

Conclusion

It is clear that social networking is taking place among women and couples seeking abortion in restrictive settings, although to varying degrees depending on the context. A better understanding of information-gathering behaviours could inform the development of programmes that can provide accurate information about safer abortion options. Regarding adolescent contraceptive use, recent reviews shows that peer-led interventions were poor at changing behaviours, but succeeded in conveying information and increasing reproductive health knowledge [7, 99]. As access to the right information is key to obtain a safe abortion in a restrictive setting, [59] there may be some scope for peer-led interventions to improve access to legal and safe abortion care. Also, given the evidence to show that social networking exists among women who seek abortions in restricted settings, this network could be utilized to inform future sampling strategies for outlining the sociodemographic characteristics of abortion seekers and safety of abortions in these settings. Conversely, these results also underline that network sampling methods are not suitable for documenting abortion practices in a representative manner in restrictive low and middle-income countries where a large share of women access abortion services anonymously, without the help of network members beyond their intimate circle.

Page 12 of 15

Abbreviations

LMICs: Low- and middle-income countries.

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s12978-021-01165-0.

Additional file 1: Search strategy.

Acknowledgements

This publication received support from UNDP-UNFPA-UNICEF-WHO-World Bank Special Programme of Research, Development and Research Training in Human Reproduction (HRP), World Health Organization. It also received support from the Swiss National Centre of Competence in Research LIVES – Overcoming vulnerability: Life course perspectives (NCCR LIVES), which is financed by the Swiss National Science Foundation (grant number: 51NF40-160590). The authors are grateful to WHO and the Swiss National Science Foundation for their financial assistance.

Disclaimer

The authors alone are responsible for the views expressed in this article and they do not necessarily represent the views, decisions or policies of the institutions with which they are affiliated.

Authors' contributions

CR, BG, and CK designed the study; AM performed the search; AM extracted the data and CR reviewed the extraction; CR and AM analyzed the data; AM wrote the methods section and CR the rest. All authors read and approved the final manuscript.

Funding

Not applicable

Availability of data and materials

Not applicable.

Declarations

Ethics approval and consent to participate Not applicable.

Consent for publication

Not applicable.

Competing interests

Not applicable.

Author details

¹Université de Genève, Geneva, Switzerland. ²Institut National d'Etudes Démographiques, Paris, France. ³University of Colorado, Denver, USA. ⁴UNDP-UNFPA-UNICEF-WHO-World Bank Special Programme of Research, Development and Research Training in Human Reproduction (HRP), World Health Organization, Geneva, Switzerland.

Received: 24 August 2020 Accepted: 23 May 2021 Published online: 07 June 2021

References

- 1. Ehsan A, Klaas HS, Bastianen A, Spini D. Social capital and health: a systematic review of systematic reviews. SSM Popul Health. 2019;8:100425.
- Portes A. Social capital: its origins and applications in modern sociology. Ann Rev Sociol. 1998;24(1):1–24.
- 3. Rossier C, Bernardi L. Social interaction effects on fertility: Intentions and behaviors. Eur J Population. 2009;25(4):467–85.

- 4. Cissé S, Sauvain-Dugerdil C, Rossier C. Family configurations and maternal healthcare seeking in Bamako. J Popul Econ. submitted.
- Rutenberg N, Watkins SC. The buzz outside the clinics: conversations and contraception in Nyanza Province, Kenya. Studies Family Planning. 1997;28:290–307.
- Bellizzi S, Sobel HL, Obara H, Temmerman M. Underuse of modern methods of contraception: underlying causes and consequent undesired pregnancies in 35 low-and middle-income countries. Hum Reprod. 2015;30(4):973–86.
- Kim CR, Free C. Recent evaluations of the peer-led approach in adolescent sexual health education: a systematic review. Perspect Sex Reprod Health. 2008;40(3):144–51.
- Johnson BR Jr, Mishra V, Lavelanet AF, Khosla R, Ganatra B. A global database of abortion laws, policies, health standards and guidelines. Bull World Health Organ. 2017;95(7):542.
- Cook RJ, Dickens BM. Reducing stigma in reproductive health. Int J Gynecol Obstet. 2014;125(1):89–92.
- Ganatra B, Gerdts C, Rossier C, Johnson BR Jr, Tunçalp Ö, Assifi A, et al. Global, regional, and subregional classification of abortions by safety, 2010–14: estimates from a Bayesian hierarchical model. Lancet. 2017;390(10110):2372–81.
- Singh S, Remez L, Sedgh G, Kwok L, Onda T. Abortion worldwide 2017: uneven Progress and unequal AccessAbortion worldwide 2017: uneven Progress and unequal Access. 2018.
- Guillaume A, Rossier C. Abortion around the world. An overview of legislation, measures, trends, and consequences. Population. 2018;73(2):217–306.
- Howell NL. The search for an abortionist. Chicago: University of Chicago Press; 1969.
- Rossier C. Abortion: an open secret? Abortion and social network involvement in Burkina Faso. Reprod Health Matters. 2007;15(30):230–8.
- 15. Simmel G. Friendship, love and secrecy. Am J Sociol. 1906;11:457-64.
- Simmel G. The secret and the secret society. In: Wolff K, editor. The Sociology of Georg Simmel. New York: Free Press; 1950.
- Astbury-Ward E, Parry O, Carnwell R. Stigma, abortion, and disclosure findings from a qualitative study. J Sex Med. 2012;9(12):3137–47.
- Batya E. Induced abortion in Rajasthan, India: Methodological and substantive issues. Baltimore: The Johns Hopkins University; 2005.
- Butts CT. Social network analysis: a methodological introduction. Asian J Soc Psychol. 2008;11(1):13–41.
- Feehan DM, Umubyeyi A, Mahy M, Hladik W, Salganik MJ. Quantity versus quality: a survey experiment to improve the network scale-up method. Am J Epidemiol. 2016;183(8):747–57.
- Petitet PH, Ith L, Cockroft M, Delvaux T. Towards safe abortion access: an exploratory study of medical abortion in Cambodia. Reprod Health Matters. 2014;22:47–55.
- 22. Silveira P, McCallum C, Menezes G. Personal experiences with induced abortions in private clinics in Northeast Brazil. Cadernos de saude publica. 2016. https://doi.org/10.1590/0102-311x00004815.
- Kebede MT, Middelthon A-L, Hilden PK. Negotiating the social and medical dangers of abortion in Addis Ababa: an exploration of young, unmarried women's abortion-seeking journeys. Health Care Women Int. 2018;39(2):186–207.
- 24. Kebede MT, Hilden PK, Middelthon AL. The tale of the hearts: deciding on abortion in Ethiopia. Cult Health Sex. 2012;14(4):393–405.
- Brack CE, Rochat RW, Bernal OA. "It's a Race Against the Clock": a qualitative analysis of barriers to legal abortion in Bogotá, Colombia. Int Perspect Sex Reprod Health. 2017;43(4):173–82.
- DePiñeres T, Raifman S, Mora M, Villarreal C, Foster DG, Gerdts C. 'I felt the world crash down on me': women's experiences being denied legal abortion in Colombia. Reprod Health. 2017;14(1):1–9.
- Kalyanwala S, Jejeebhoy SJ, Francis Zavier A, Kumar R. Experiences of unmarried young abortion-seekers in Bihar and Jharkhand, India. Cult Health Sex. 2012;14(3):241–55.
- Jejeebhoy SJ, Kalyanwala S, Zavier AF, Kumar R, Jha N. Experience seeking abortion among unmarried young women in Bihar and Jharkhand, India: delays and disadvantages. Reprod Health Matters. 2010;18(35):163–74.

- Banerjee SK, Andersen KL, Buchanan RM, Warvadekar J. Woman-centered research on access to safe abortion services and implications for behavioral change communication interventions: a cross-sectional study of women in Bihar and Jharkhand, India. BMC Public Health. 2012;12(1):175.
- Elul B, Bracken H, Verma S, Ved R, Singhi R, Lockwood K. Unwanted pregnancy and induced abortion in Rajasthan, India: a qualitative exploration. Population Council: New Delhi; 2004.
- Behera D, Bharat S, Gawde NC. Induced abortion practices in an urban Indian slum: exploring reasons, pathways and experiences. J Family Reprod Health. 2015;9(3):129.
- 32. Banerjee SK, Andersen K. Exploring the pathways of unsafe abortion in Madhya Pradesh. India Global Public Health. 2012;7(8):882–96.
- Banerjee SK, Andersen KL, Warvadekar J. Pathways and consequences of unsafe abortion: a comparison among women with complications after induced and spontaneous abortions in Madhya Pradesh, India. Int J Gynecol Obstet. 2012;118:S113–20.
- 34. Banerjee SK, Andersen KL, Warvadekar J, Aich P, Rawat A, Upadhyay B. How prepared are young, rural women in India to address their sexual and reproductive health needs? A cross sectional assessment of youth in Jharkhand. Reprod Health. 2015;12(1):1–10.
- Elul B. Determinants of induced abortion: an analysis of individual, household and contextual factors in Rajasthan. India J Biosocial Sci. 2011;43(1):1–17.
- Zamanian M, Baneshi MR, Haghdoost A, Zolala F. Estimating the visibility rate of abortion: a case study of Kerman, Iran. BMJ open. 2016;6(10):e012761.
- Orner P, de Bruyn M, Cooper D.'It hurts, but I don't have a choice, I'm not working and I'm sick': decisions and experiences regarding abortion of women living with HIV in Cape Town, South Africa. Cult Health Sex. 2011;13(7):781–95.
- Orner P, De Bruyn M, Harries J, Cooper D. A qualitative exploration of HIVpositive pregnant women's decision-making regarding abortion in Cape Town, South Africa. SAHARA-J. 2010;7(2):44–51.
- Gerdts C, Raifman S, Daskilewicz K, Momberg M, Roberts S, Harries J. Women's experiences seeking informal sector abortion services in Cape Town, South Africa: a descriptive study. BMC Womens Health. 2017;17(1):95.
- Harries J, Orner P, Gabriel M, Mitchell E. Delays in seeking an abortion until the second trimester: a qualitative study in South Africa. Reprod Health. 2007;4(1):1–8.
- Ojha N, Bista KD. Situation analysis of patients attending TU Teaching Hospital after medical abortion with problems and complications. J Nepal Med Assoc. 2013. https://doi.org/10.31729/jnma.2274.
- Andersen KL, Khanal RC, Teixeira A, Neupane S, Sharma S, Acre VN, et al. Marital status and abortion among young women in Rupandehi. Nepal BMC women's health. 2015;15(1):17.
- Puri M, Vohra D, Gerdts C, Foster DG. "I need to terminate this pregnancy even if it will take my life": a qualitative study of the effect of being denied legal abortion on women's lives in Nepal. BMC Womens Health. 2015;15(1):1–11.
- 44. Rocca CH, Puri M, Dulal B, Bajracharya L, Harper C, Blum M, et al. Unsafe abortion after legalisation in Nepal: a cross-sectional study of women presenting to hospitals. BJOG. 2013;120(9):1075–84.
- Messinger CJ, Mahmud I, Kanan S, Jahangir YT, Sarker M, Rashid SF. Utilization of mobile phones for accessing menstrual regulation services among low-income women in Bangladesh: a qualitative analysis. Reprod Health. 2017;14(1):7.
- 46. Alemayehu M, Yebyo H, Medhanyie AA, Bayray A, Fantahun M, Goba GK. Determinants of repeated abortion among women of reproductive age attending health facilities in Northern Ethiopia: a case–control study. BMC Public Health. 2017;17(1):188.
- Bury L, Aliaga Bruch S, Machicao Barbery X, Garcia PF. Hidden realities: what women do when they want to terminate an unwanted pregnancy in Bolivia. Int J Gynecol Obstet. 2012;118:S4–9.
- Jayaweera RT, Ngui FM, Hall KS, Gerdts C. Women's experiences with unplanned pregnancy and abortion in Kenya: a qualitative study. PloS one. 2018;13(1):e0191412.

- Alsibiani SA. Use of misoprostol for self-induced medical abortions among Saudi women: a call for attention. Gynecol Obstet Invest. 2014;78(2):88–93.
- 50. Arambepola C, Rajapaksa LC. Decision making on unsafe abortions in Sri Lanka: a case-control study. Reprod Health. 2014;11(1):91.
- 51. Arilha MM. Misoprostol: Pathways, Mediation And Social Networks For Access To Abortion Using Medication In The Context Of Illegality In The State Of Sao Paulo (misoprostol: Percursos, Mediações E Redes Sociais Para O Acesso Ao Aborto Medicamentoso Em Contextos De Ilegalidade No Estado De São Paulo). Ciencia e Saude Coletiva. 2012.
- Diniz D, Medeiros M. Itineraries and methods of illegal abortion in five Brazilian state capitals. Ciencia & saude coletiva. 2012;17(7):1671.
- 53. Madeiro AP, Diniz D. Induced abortion among Brazilian female sex workers: a qualitative study. Cien Saude Colet. 2015;20:587–93.
- Heilborn ML, Cabral CDS, Brandao ER, Faro L, Cordeiro F, Azize RL. Abortion itineraries in a clandestine context in the city of Rio de Janeiro-Brazil. Ciencia saude coletiva. 2012;17(7):1699–708.
- 55. Casas L, Vivaldi L. Abortion in Chile: the practice under a restrictive regime. Reprod Health Matters. 2014;22(44):70–81.
- Manríquez IP, Standen CM, Carimoney AÁ, Richards A. Experience of clandestine use of medical abortion among university students in Chile: a qualitative study. Contraception. 2018;97(2):100–7.
- Ramos S, Romero M, Aizenberg L. Women's experiences with the use of medical abortion in a legally restricted context: the case of Argentina. Reprod Health Matters. 2014;22(sup44):4–15.
- Haaland ME, Haukanes H, Zulu JM, Moland KM, Blystad A. Silent politics and unknown numbers: Rural health bureaucrats and Zambian abortion policy. Social Sci Med. 2020;251:112909.
- 59. Coast E, Murray SF. "These things are dangerous": understanding induced abortion trajectories in urban Zambia. Soc Sci Med. 2016;153:201–9.
- 60. Freeman E, Coast E, Murray SF. Men's roles in women's abortion trajectories in urban Zambia. Int Perspect Sex Reprod Health. 2017;43(2):89–98.
- Cresswell JA, Schroeder R, Dennis M, Owolabi O, Vwalika B, Musheke M, et al. Women's knowledge and attitudes surrounding abortion in Zambia: a cross-sectional survey across three provinces. BMJ open. 2016;6(3):e010076.
- Dahlbäck E, Maimbolwa M, Kasonka L, Bergström S, Ransjö-Arvidson AB. Unsafe induced abortions among adolescent girls in Lusaka. Health Care Women Int. 2007;28(7):654–76.
- Izugbara CO, Egesa C, Okelo R. High profile health facilities can add to your trouble': women, stigma and un/safe abortion in Kenya. Social Sci Med. 2015;141:9–18.
- Osur J, Orago A, Mwanzo I, Bukusi E. Social networks and decision making for clandestine unsafe abortions: evidence from Kenya. Afr J Reprod Health. 2015;19(1):34–43.
- 65. Izugbara CO, Otsola KJ, Ezeh AC. (), Men, women, and abortion in central Kenya: a study of lay narratives. Med Anthropol. 2009;28(4):397–425.
- Mitchell EM, Halpern CT, Kamathi EM, Owino S. Social scripts and stark realities: Kenyan adolescents' abortion discourse. Culture Health Sexuality. 2006;8(6):515–28.
- Penfold S, Wendot S, Nafula I, Footman K. A qualitative study of safe abortion and post-abortion family planning service experiences of women attending private facilities in Kenya. Reprod Health. 2018;15(1):1–8.
- 68. Hill ZE, Tawiah-Agyemang C, Kirkwood B. The context of informal abortions in rural Ghana. J Womens Health. 2009;18(12):2017–22.
- Kumi-Kyereme A, Gbagbo FY, Amo-Adjei J. Role-players in abortion decision-making in the Accra Metropolis, Ghana. Reproductive health. 2014;11(1):70.
- Aniteye P, Mayhew, S. Attitudes and experiences of women admitted to hospital with abortion complications in Ghana. Afr J Reprod Health. 2011;15(1).
- Appiah-Agyekum NN. Medical abortions among university students in Ghana: implications for reproductive health education and management. Int J Women's Health. 2018;10:515.
- Ganle JK, Obeng B, Yeboah JY, Tagoe-Darko E, Mensah CM. Disparities in abortion experience and access to safe abortion services in Ghana: Evidence from a retrospective survey. Afr J Reprod Health. 2016;20(2):43–52.

- 73. Rominski SD, Lori JR, Morhe ES. "My friend who bought it for me, she has had an abortion before." The influence of Ghanaian women's social networks in determining the pathway to induced abortion. J Family Planning Reprod Health Care. 2017;43(3):216–21.
- Rominski SD, Darteh E, Dickson KS, Munro-Kramer M. Attitudes toward abortion among students at the University of Cape Coast. Ghana Sexual Reproductive Healthcare. 2017;11:53–9.
- Nyanzi S, Nyanzi B, Bessie K. " Abortion? That's for women!" Narratives and experiences of commercial motorbike riders in south-western Uganda. Afr J Reprod Health. 2005;9(1):142–61.
- Atuyambe LM, Kibira SP, Bukenya J, Muhumuza C, Apolot RR, Mulogo E. Understanding sexual and reproductive health needs of adolescents: evidence from a formative evaluation in Wakiso district. Uganda Reproductive health. 2015;12(1):1–10.
- Marlow HM, Shellenberg K, Yegon E. Abortion services for sex workers in Uganda: successful strategies in an urban clinic. Cult Health Sex. 2014;16(8):931–43.
- Norris A, Harrington BJ, Grossman D, Hemed M, Hindin MJ. Abortion experiences among Zanzibari women: a chain-referral sampling study. Reprod Health. 2016;13(1):23.
- Plummer ML, Wamoyi J, Nyalali K, Mshana G, Shigongo ZS, Ross DA, Wight D. Aborting and suspending pregnancy in rural Tanzania: an ethnography of young people's beliefs and practices. Stud Fam Plann. 2008;39(4):281–92.
- Rasch V, Kipingili R. Unsafe abortion in urban and rural Tanzania: method, provider and consequences. Tropical Med Int Health. 2009;14(9):1128–33.
- Rouhani SA, Scott J, Burkhardt G, Onyango MA, Haider S, Greiner A, Bartels SA. A quantitative assessment of termination of sexual violence-related pregnancies in eastern Democratic Republic of Congo. Conflict Health. 2016;10(1):9.
- Burkhardt G, Scott J, Onyango MA, Rouhani S, Haider S, Greiner A, Bartels S. Sexual violence-related pregnancies in eastern Democratic Republic of Congo: a qualitative analysis of access to pregnancy termination services. Conflict Health. 2016;10(1):30.
- Ouédraogo R, Sundby J. Social determinants and access to induced abortion in Burkina Faso: from two case studies. Obstetrics Gynecol Int. 2014. https://doi.org/10.1155/2014/402456.
- Baxerres C, Boko I, Konkobo A, Ouattara F, Guillaume A. Abortion in two francophone African countries: a study of whether women have begun to use misoprostol in Benin and Burkina Faso. Contraception. 2018;97(2):130–6.
- Pourette D, Mattern C, Ratovoson R, Raharimalala P. Complications with use of misoprostol for abortion in Madagascar: between ease of access and lack of information. Contraception. 2018;97(2):116–21.
- Smith SS. The challenges procuring of safe abortion care in Botswana. Afr J Reprod Health. 2013;17(4):43–55.
- 87. Gipson JD, Hirz AE, Avila JL. Perceptions and practices of illegal abortion among urban young adults in the Philippines: a qualitative study. Stud Fam Plann. 2011;42(4):261–72.
- Arnott G, Tho E, Guroong N, Foster AM. To be, or not to be, referred: a qualitative study of women from Burma's access to legal abortion care in Thailand. PloS one. 2017;12(6):0179365.
- Tousaw E, La RK, Arnott G, Chinthakanan O, Foster AM. "Without this program, women can lose their lives": migrant women's experiences with the Safe Abortion Referral Programme in Chiang Mai, Thailand. Reprod Health Matters. 2017;25(51):58–68.
- Tousaw E, Moo SNHG, Arnott G, Foster AM. "It is just like having a period with back pain": exploring women's experiences with community-based distribution of misoprostol for early abortion on the Thailand-Burma border. Contraception. 2018;97(2):122–9.
- Grossman D, Garcia SG, Kingston J, Schweikert S. Mexican women seeking safe abortion services in San Diego, California. Health Care Women Int. 2012;33(11):1060–9.
- Tong WT, Low WY, Wong YL, Choong SP, Jegasothy R. Exploring pregnancy termination experiences and needs among Malaysian women: A qualitative study. BMC Public Health. 2012. 12(1).
- 93. Alubo O. Adolescent reproductive health practices in Nigeria. Afr J Reprod Health. 2001;5(3):109–19.

- Bankole A, Sedgh G, Oye-Adeniran BA, Adewole IF, Hussain R, Singh S. Abortion-seeking behaviour among Nigerian women. J Biosoc Sci. 2008;40(2):247.
- Jackson E, Johnson BR, Gebreselassie H, Kangaude GD, Mhango C. A strategic assessment of unsafe abortion in Malawi. Reprod Health Matters. 2011;19(37):133–43.
- Hess RF. Women's stories of abortion in southern Gabon. Africa J Transcult Nursing. 2007;18(1):41–8.
- Albuja LD, Cianelli R, Anglade D, Owusu B, Joseph L, Sailsman S, Ferrer L. Women's and healthcare workers' beliefs and experiences surrounding abortion: the case of Haiti. J Nurs Scholarsh. 2017;49(2):170–6.
- Shahawy S, Diamond MB. Perspectives on induced abortion among Palestinian women: religion, culture and access in the occupied Palestinian territories. Cult Health Sex. 2018;20(3):289–305.
- Fikree FF, Abshiro WK, Mai MM, Hagos KL, Asnake M. The effect of peer education in dispelling myths and misconceptions about long-acting reversible contraception among Ethiopian youth. Afr J Reprod Health. 2018;22(3):90–9.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

